

We would like to thank the reviewers and the editor-in-chief for the critical evaluation of the manuscript and helpful suggestions. We are thankful to the reviewers for finding our editorial good. We believe that the issues raised are important, and we are thankful to the reviewers for their insightful comments and criticisms. We have clarified issues raised by reviewers and believe that addressing the criticisms has significantly improved the manuscript. The responses to the specific comments for the reviewer are given below:

Responses to Reviewer 1

Comment (1) The language classification is Grade B. Please visit the following website for the professional English language editing companies that we recommend: <https://www.wjgnet.com/bpg/gerinfo/240>.

Response: We thank the reviewer for the comment. We have used a professional English language editing service and revised the manuscript accordingly.

Comment (2) The “Key Words” does not meet the requirements: The first letter of each keyword will be capitalized, and each keyword will be separated by a semicolon, with no terminal period. An example of correct formatting is: Non-alcoholic fatty liver disease; Alcoholic liver disease; Non-alcoholic steatohepatitis; Insulin resistance; Oxidative stress.

Response: We thank the reviewer for the comment. We have revised the keywords as per the suggestion. Please refer to the revised version of the manuscript.

Comment (3) **Please revise the reference formatting.** The format of in-text citation of references should be [References Number], which should be with no space between “[]” and the preceding word. **Example:** The pathophysiology is thought to be due to an increased arterial flow that leads to secondary hepatocellular hyperplasia[1,2].

To ensure the accuracy of the references, please use "**Edit References by Auto-Analyser**" (<https://www.f6publishing.com/Forms/main/ArticleReferenceTool.aspx>) to edit the references of the manuscript.

Response: We thank the reviewer for the comment. We have revised the references as per the guideline of journal and using the Auto-Analyser. Please refer to the revised version of the manuscript.

Comment (4) The article which this editorial discussed has been listed in the references list (Ref. 21). Please quote in the correct format <Liu YJ, Zou C, Wu YY. Metastatic clear cell sarcoma of the pancreas: A rare case report. World J Clin Cases 2024, 12: 1448-1453. DOI: 10.12998/wjcc.v12.i8.1448 PMID: 38576799>

Response: We thank the reviewer for the comment. We have revised the mentioned reference as per the reviewer's suggestion and the Auto-Analyser. Please refer to the revised version of the manuscript.

Comment (5) Please provide the filled conflict-of-interest disclosure form.

Response: We thank the reviewer for the comment. We have uploaded the "Conflict-of-interest" disclosure form in the resubmission. Please find the file.